

Delft University of Technology

Stakeholder Perspectives on Community Energy Contributing to the Use of Renewable Energy Sources and Improving Energy Security in Nigeria

Ogunleye, O.S.; Coenen, F.; Hoppe, T.

DOI 10.3390/en15197390

Publication date 2022 **Document Version** Final published version Published in

Energies

Citation (APA)

Ogunleye, O. S., Coenen, F., & Hoppe, T. (2022). Stakeholder Perspectives on Community Energy Contributing to the Use of Renewable Energy Sources and Improving Energy Security in Nigeria. *Energies*, *15*(19), 1-25. Article 7390. https://doi.org/10.3390/en15197390

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.





Article Stakeholder Perspectives on Community Energy Contributing to the Use of Renewable Energy Sources and Improving Energy Security in Nigeria

Olaoluwa Sunday Ogunleye¹, Frans Coenen¹ and Thomas Hoppe^{2,*}

- ¹ Twente Centre for Studies in Technology and Sustainable Development (CSTM), University of Twente, 7522 NB Enschede, The Netherlands
- ² Organisation and Governance (OG), Department of Multi-Actor Systems (MAS), Faculty of Technology, Policy and Management (TPM), Delft University of Technology, 2628 BX Delft, The Netherlands
- * Correspondence: t.hoppe@tudelft.nl

Abstract: The need for improved energy security in Nigeria cannot be over-emphasized. Currently, energy security is rather poor, while access to energy is fundamental to socio-economic development and poverty alleviation. Renewable energy could potentially contribute to resolving this because renewable sources such as solar radiation are more available and sustainable, and can be set up in small generation units, meaning that it is suitable for community management and ownership. In theory, a community energy approach could well apply. In this paper, the main research question is: In what ways can community energy initiatives contribute to increasing the use of renewable energy sources and improving energy security in Nigeria according to selected stakeholders and households? A mixed methods research approach was used to answer this question, with stakeholder interviews and survey data from 124 residents in two case studies of selected housing estates in Lagos. The results show that 58% of the households and most of the stakeholders express support for community renewable energy as a viable approach for increasing energy access and greening energy supply. The present study shows that there is a need to raise awareness and support projects for effective and supportive renewable energy policy to encourage local renewable energy community formation.

Keywords: renewable energy; community energy; energy security; renewable energy technology; energy policy; microgrid

1. Introduction

Renewable energy is used in many countries across the world. It can potentially provide reliable and clean power to low-income households in both rural and urban areas. According to UNEP [1] renewable energy can promote energy security by decentralizing energy supply with micro, mini, modular, and rapidly deployable energy projects that are particularly suited to the electrification of rural communities in developing countries. Here, energy security is defined as the uninterrupted availability of energy sources at an affordable price [2].

Over the past four decades, the attention of world leaders has shifted from nonrenewable energy sources to clean energy sources that are environmentally sustainable and economically viable [3]. For these reasons diversifying energy sources and adopting a greater share of renewable energy sources is also expected in Nigeria, where it could be part of the national strategy, which indicates the country's commitment to global climate change agreements and could help to remedy the electricity crisis referring to the failure of the Nigerian power sector to provide adequate electricity supply to domestic households and industrial producers despite having a rapidly growing economy.

Currently, Nigeria is the largest producer of crude oil in sub-Saharan Africa. Most of the energy supply in the country comes from fossil fuels. Natural gas is used for thermal



Citation: Ogunleye, O.S.; Coenen, F.; Hoppe, T. Stakeholder Perspectives on Community Energy Contributing to the Use of Renewable Energy Sources and Improving Energy Security in Nigeria. *Energies* **2022**, *15*, 7390. https://doi.org/10.3390/ en15197390

Academic Editor: Seung-Hoon Yoo

Received: 9 September 2022 Accepted: 29 September 2022 Published: 8 October 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). energy production. The majority of Nigeria's electricity supplied via the central electricity grid is generated by using hydropower and natural gas. The enactment of the Electric power reform act in 2005 allowed for renewable energy, such as solar energy, into the energy mix [4].

Although the Nigerian government made a financial commitment to the power sector, in terms of output, no significant improvements were made to increase electricity generation despite vast economic and demographic growth in the country. In some cases, such as Lagos and Ogun State, power supply systems even deteriorated, more particularly with regard to transmission and distribution network cables. The national electricity grid is infamous for experiencing total shutdowns. In February, May, July, and August 2021, and in March 2022, it collapsed, which led to a nationwide blackout, causing a myriad of problems in society [5], such as increased unemployment, poverty, slow economic growth, all related to power failure [6].

Due to the problematic nature of the electricity supply from the central grid, households and industries have become keen to resolve power supply issues themselves by using diesel- or petrol-powered generators for their own energy need [7]. Individual homes remote from the electricity grid use fossil fuels to generate heat and electricity supply (i.e., natural gas or oil). In case of short supply, diesel and petrol generators are used. Fossil-fuel-generated electricity (i.e., using coal, natural gas, or oil) is used in the industry.

1.1. Community Energy

If renewable energy is to be used to a greater extent, it is important that it also resonates at different territorial levels, including the local one. When applied at this level, renewable energy sources can become part of community systems, which comprise technical, social, and institutional components that co-exist and co-evolve. Theoretically, this refers to re-organizing local energy systems, allowing the simultaneous integration of renewable distributed energy resources and involving local communities [8,9]. Here, the concept of community energy is key. Community energy can be defined in different ways [10] but commonly refers to a wide range of citizen initiatives that are concerned with greening (parts of) energy systems, usually but not only on a local or regional scale [11] by collectively organizing decentralized energy actions, with application in the domains of renewable energy, energy efficiency, and energy service provision [12]. Community energy can be viewed as both a bottom-up, grassroots innovation [13] and as a social innovation [14]. Community energy system development can be seen as a potentially important strategy that can be adopted socially and by policymakers to contribute to energy and climate (mitigation) policy goals [14,15]. This does, however, require households, entrepreneurs, industries, and local governments to become motivated to actively participate in or empower community energy systems [16,17].

The drive towards climate change mitigation and the opportunity to empower consumers in the developed and developing worlds points increasingly towards moving away from centralized energy supply models [18]. This would call for the increased use of decentralized models, using distributed generation, which would require allowing community renewable energy projects (including households as energy producers, i.e., 'prosumers') to access and feed excess distributed generated energy into distribution grids [19–21]. This, in turn, would call for institutional change, removing or adapting current legislation that prevents feed-in of distributed generation [22,23] and formulating and implementing economic incentives, regulations, and 'game changers' that encourage and mainstream it [24]. To improve distributed generation and energy access, and empower community energy in Nigeria, theoretically speaking, the central government could initiate and enact this into national and regional legislation.

1.2. Current Electricity Regulatory Framework in Nigeria and Future Plans

Based on experiences with previously failed reforms, the 2010 administration initiated a power sector reform plan, which aimed at generating 40,000 MW of power by 2020 [25]

whilst expanding generation and transmission capacity and improving the electricity market structure. The reform process was successful in handing over the power plants to private sector business companies using tenders and auctions [26,27].

However, transferring the distribution and generation networks to the new owners created a new quasi-monopoly in the electricity sector that eventually even caused the failure of the reform [28]. In hindsight, this was predictable due to the decision-making process that was shrouded in secrecy between the federal authorities and the incumbent electricity companies' networks and without public participation. This led to the failure of the reform, and the government projection of 40,000 MW was unsuccessful. This contrasts with decision-making in some of the developed world countries, having fairly responsive and democratic governance systems in place. Public participation has helped shape reforms in the energy sector, with governments paying attention to ideas from the public while acknowledging key public values [29].

As of 2022, many electricity distribution infrastructures in Nigeria are obsolete. Distributed system operators have failed to invest in revamping these infrastructures, which has hampered consumers' ability to access the grid. The current government recently tried to revamp the power sector by renegotiating the previous government's contract with the generation and distribution companies. However, this failed due to the contractual clause. Another critical point in the contract is that Nigeria's system cannot take more than 4000 MW of electricity to avoid collapse. Critics argue that the national government should have improved the system's capacity before signing a poor contract [30].

In July 2019, the Federal Government of Nigeria set up a Presidential Power Initiative to strategize on how to improve the country's installed capacity and reduce power loss [29]. Currently, Nigeria has a total of 13 GW installed capacity, but only about 3.4 GW is available for distribution to consumers [31].

The current energy regulatory framework in Nigeria is based on the Electrical Power Sector Act (EPSR), which was passed by the National Assembly in 2005 and repealed by the National Electric Power Authority Act (NEPA). The enactment of the EPSR opened the Nigeria electricity sector to private sector investment by breaking the previous monopoly of NEPA. In succession to the EPSRA 2005, the state-owned vertically integrated utility underwent partial unbundling, which led to the creation of six electricity generation companies (i.e., Gencos). This pertained to eleven distribution companies that were privatized in 2013. In the meanwhile, the federal government retained one hundred percent ownership of the national transmission system operator [26].

To further develop the energy sector, the government introduced the Nigerian Renewable Energy and Energy Efficiency Policy in 2015 (NREEEP) intending to inject more hydropower, biomass, solar, wind, and geothermal energy into the energy mix through the distributed energy system, whilst implementing a few economic incentives, such as netmetering, tax credit, and a feed-in tariff that would promote renewable energy production projects in Nigeria's capital, Lagos. The NREEEP was intended to promote national energy security and an efficient energy delivery system that is cost-effective and would work in an environmentally sustainable manner across multiple economic sectors.

NREEP can be viewed as a veritable policy that could promote the energy transition effort of Nigeria and potentially also the development of local community energy initiatives. However, the implementation of NREEP was troublesome because Nigeria did not have an overarching renewable energy regulatory framework in place to enforce compliance with the policy. Moreover, there is a lack of organization to implement it [27,30].

Over the last five years, the federal government of Nigeria has made efforts to improve the grid capacity and reduce the incessant collapse of the power system. The government of Nigeria signed an agreement with Siemens in 2018 to help deliver 25,000 MW by 2025 under the Presidential Power Initiative. The project entailed three phases. The first phase began in 2021 for ten months which would help push the installed capacity to 7000 MW. The second phase would raise the installed capacity to 11,000 MW, and the third phase would increase the installed capacity to 25,000 MW. The phases were designed to cover the expansion of the transmission and distribution grids, improve access to affordable, efficient, reliable electricity and therefore increase overall increase energy security.

On 24 August 2022, the government of Nigeria launched Nigeria's national Energy Transition Plan (NETP), which covers power, transport, cooking, and industry to reduce emissions and power national economic development. NETP intends to create investment opportunities in solar energy, hydrogen storage, and electric vehicles that would create 330,000 jobs by 2030 and 840,000 jobs by 2060 in the power and transportation sectors. One of the key objectives of NETP was to let go of the diesel and petrol generators, which account for the current bulk of power generation capacity and CO₂ emission in the country. NETP also sought to expand the generation of gas generation capacity to meet the electricity demand and integrate renewable energy.

Facing major problems related to the centralized electricity supply system, one may wonder whether alternative models to electricity supply can be developed and considered for utilization. One of these models refers to community renewable energy initiatives, using distributed generation, giving households and firms opportunities to achieve better access to electricity, and enjoy a higher degree of energy security whilst using energy in a more environmentally friendly supply system. However, currently, there is no incentive to promote community renewable energy initiatives, for it is currently part of any formal policy framework, despite the establishment of the NREEEP [25].

1.3. Community Energy as a Means to Improve Energy Security

The integration of renewable energy into the energy mix has grown over the past two decades to ensure the security of the supply of energy, particularly concerning supplying energy in remote areas and, in the case of solar PV, ensuring electricity supply in times of undersupply by the central electricity grid. However, in the current situation, the central electricity grid does not adequately supply electricity to housing estates. There are lots of blackouts, and the security of the supply cannot be guaranteed. It is fair to state that the centralized nature of the electricity market in Nigeria has created energy insecurity across the country, with problems particularly manifesting in populated areas such as Lagos State, which is especially problematic because it is also the commercial hub of the country.

Community renewable energy could potentially help reduce dependency on the centralized electricity grid. This can be achieved by creating more distributed generation, particularly by using renewable sources, such as solar radiation, and in such a way that end-users have better access to electricity. One way of doing so is by developing community energy systems at the local level [8], which are characterized by community-owned assets, such as distributed generation (e.g., solar PV installations), but also energy storage and energy infrastructure, and can be designed to operate as a microgrid (either connected or disconnected to the central grid), or theoretically even in an islanded mode [31]. In experimental settings and demonstration pilots in remote areas, this concept has already been tested [32,33]. Within a community energy system operating such microgrids, members of a housing estate can theoretically become 'prosumers' of decentralized produced electricity and collaborate within a renewable energy initiative (organization) to control local electricity systems (e.g., as a microgrid).

Renewable energy communities—often organized in the form of a citizen energy cooperative [34]—can address the social, economic, and organizational side of the system by raising awareness of social acceptance, running operations, and controlling the systems (although oftentimes outsourced to an energy service company; 'ESCO') [35]. In this way, housing estates could potentially become more resilient and increase residents' confidence about coping with blackouts, outages, or the undersupply of electricity from the central grid whilst increasing the availability, accessibility, and possibly even the affordability of electricity. In addition, such systems also offer benefits in terms of wellbeing—especially in remote areas—because they replace inferior and health-threatening options, such as kerosene lamps, candles, and in some cases, even darkness [36].

1.4. Research Aim and Question

In the present paper, the focus is on community energy as an alternative model to encourage the production and consumption of renewable energy and, by doing so, help local communities to achieve higher levels of energy security, especially in times of poor electricity supply from the central grid. This is not to encourage or advocate replacing the current centrally-led electricity system in Nigeria but to explore perceptions by stakeholders about developing and running local community energy systems complementing the central system in times of need to overcome supply problems that go along with the former. The locus of this study is two housing estates in Nigeria's capital, Lagos. The main research question of this paper is: In what ways can community renewable energy initiatives contribute to increasing the use of renewable energy sources and improving energy security in Nigeria according to selected stakeholders and households?

This research aims to contribute knowledge towards the possibilities of stimulating local communities to formulate community renewable energy initiatives, even as the government has reiterated its commitment to adding 4000 MW of electricity to the grid every year. If taken seriously, we presume that community renewable energy could contribute considerably to achieving this figure.

This paper is structured as follows. Section 2 presents a literature review. Section 3 presents the research design and methodology. Section 4 presents the results. Finally, in Section 5, the results are discussed on their academic merit, and Section 6 presents the conclusions and limitations, as well as suggestions for future research.

2. Literature Review

2.1. Community Renewable Energy Initiatives

Access to affordable, modern energy is a prerequisite for sustainable development and poverty alleviation and, more specifically, for achieving each of the Sustainable Development Goals (SDGs). The assumption is that social networks and relations are essential determinants of an individual's access to resources, including energy and access to knowledge and skills, funds, energy, and information [37].

Community Renewable Energy Initiatives (CREI) provide such a social structure and express the general notion that cooperation among people for a common purpose yields satisfactory results when compared to doing things only individually [38]. CREIs vary in origin, motives, resources, and approach [14]. Moreover, they are typically found in small towns, villages, or neighborhoods; they are community-driven "grassroots"; they have gone beyond single-issue actions to "managed", multi-stranded programs; and they contribute to reaching sustainability goals [39]. The initiatives are mainly organized by volunteers, who involve community members implementing projects intended to produce sustainability-related outcomes [40]. Examples of community energy projects include household energy conservation initiatives, community-run mini-buses that drive on biogas, collective solar PV plants, and community wind farms, to mention but a few [16].

An important goal CREIs often have is to achieve a just transition to low-carbon economies and societies [41]. They can be seen as a social innovation [15] that aims to disrupt traditional ownership of essential assets and change institutions of energy systems while striving to (re-)gain community ownership and democratic control of energy systems [39,42]. One of the major reasons for the formation of a CREI is to change the structure and institutions of the current centralized, organized energy system from a centralized grid system led by energy market incumbents into a distributed system using renewable energy sources with many benefits, such as improving the general wellbeing of communities, lowering carbon emissions, and contributing to reducing climate change and its effects [15].

Renewable energy production is key to most CREIs and requires having business models in place that are needed to address the ownership and control of renewable energy installations facilitated by supportive national policies and to assure the financial feasibility of the organization. Without the right business model, many CREIs will not last. However, this depends on the (supportive) policy landscape. Whereas policies that have previously supported the deployment of small-scale renewable projects are withdrawn, others are issued. This requires community energy organizations to innovate and adapt their business models if they want to succeed and survive. Nolden et al. [43] conducted a study on CREI business models in the United Kingdom and discerned three types, all heavily shaped by the policy landscape: (i) community renewable business models based on grant funding (pre-2010); (ii) small community PV projects based on feed-in tariffs (2010–2015); and (iii) large community solar PV projects based on feed-in tariffs and power purchase agreements (after 2013).

CREI projects could either be 100% owned by a community or might be developed under co-ownership arrangements with the private sector. For example, a solar park project in Ameland Island in the Netherlands was co-owned by the municipality and ENECO, an energy-supplying company [39]. Another example pertains to the solar park located in Saerbeck, Germany, with over 24,000 solar PV panels, which are wholly owned by the community [44]. Projects can involve the financing and ownership of distributed generation, with electricity generated fed into the grid rather than in addition to local use. However, this requires using a business model that assumes financial compensation, for example, by using a feed-in tariff or net-metering. To make this possible presumes that the electricity market is regulated in the sense that these instruments are readily available [17,45]. Although CREIs are mostly professed as self-deterministic and bottom-up, the initiatives greatly depend on supportive public policy [17,45–47].

While there are various models of CREIs, the motivations and ideologies behind their formation are often quite similar and may include social needs and the need to create social impact by developing localized solutions, such as educating residents, community members, juveniles, and women, whilst building more capacities [48]. More in general, the various models of community energy include cooperatives, community charity, development trust, and shares held by local organizations (See [37] for a detailed clarification of the different models) and, in most cases, adhere to the seven principles of the International Cooperative Alliance [49].

2.2. Understanding Formation of Community Renewable Energy Initiatives

An approach to understanding the formation of local community energy initiatives was developed by Boon & Dieperink [50] and later elaborated and empirically assessed by Warbroek et al. [39]. Based on these frameworks, a new one was developed to be applied in the developing country's context (See Figure 1). This framework includes an overview of stakeholders that could potentially contribute to the formation of CREIs, factors that influence the formation processes of CREIs, and various aspects to be considered, such as the three pillars of sustainability, technology, existing regulations, and socio-cultural characteristics and the market for renewable energy technologies.

Stakeholder participation, system efficiency, and the sustainability of the CREIs form the foundation of this framework. Some factors, such as household income level, awareness, and clear understanding of CREIs, can also be considered for renewable energy technology (RET) and CREIs to be developed in developing countries such as Nigeria. However, when reflecting on this, RET diffusion in Nigeria is impossible without also taking into account the regulatory frameworks, technical know-how, socio-economic-, environmental, and socio-cultural factors. These could form potential barriers to local communities initiating energy initiatives.



Figure 1. Conceptual framework CREI formation; partly adapted from [50].

As the electricity supply is not secure in Nigeria and people mostly depend on fossilfuel generators for most of their energy services, this irregular power supply in the country provides a good argument for community renewable energy initiatives in Nigerian communities. Households, local business firms, and public organizations need to control their energy system, reduce the cost of conventional fossil fuels, and reduce the menace of climate change. Therefore, CREIs provide an opportunity to offer individual and collective benefit, mostly in terms of improving energy security and sustainability.

According to the theoretical framework by Boons and Dieperink [50], there are several factors influencing the formation of CREIs. In the paragraphs below, they are described and situated in the Nigerian context.

2.2.1. Awareness Raising and Education

For society to adopt any modern technology or another type of innovation, there is a need to embark on a public enlightenment program on the benefits of renewable energy technology. Proper awareness and education [51] are the first steps in the process toward the adoption of RET or CREI. The use of plain language with less technical terms would help the public to understand how this innovative technology would be beneficial to them. Should they have proper knowledge of RET and its benefits, it will be easier to adopt and diffuse it, which would lead to the success of CREIs in communities. Awareness levels on climate change and local action on energy have increased significantly in recent years due to their potential to reduce CO_2 emissions [52].

2.2.2. Economic Factors

For a developing country such as Nigeria to adopt RET and for CREI, the initial investment cost is one of the significant factors to consider because most of the RET assets and infrastructure are not yet locally manufactured and therefore require import. In addition, there is also a lack of economic policy incentives to encourage the adoption of RET. This is even worse concerning the formation of CREIs. There is also a lack of local skilled labor to operate and maintain renewable energy assets. This could potentially hinder the adoption of RET and, subsequently, CREIs [53].

2.2.3. Regulatory Framework and Policy

Nigeria is a signatory to the Paris 2015 Climate Change Agreement. Still, the noncompliance of implementing this treaty into legislation is considered a big challenge. Corruption has been the bane of the power sector when the government is committed to implementing some of its policies; the monitoring and enforcement team in the implementing agency jeopardizes the government's efforts with corruption [54]. Regulatory barriers may arise from lengthy bureaucratic procedures and the absence of a long-term and consistent policy framework. One may generate electricity in Nigeria, but one cannot distribute it due to the lack of a sound regulatory framework to access the grid. i.e., there is currently neither a feed- tariff nor a net-metering scheme [55].

2.2.4. Environmental Factors

Nigeria is home to a diversity of natural ecosystems ranging from semi-arid savanna to mountain forests, rich seasonal floodplain environments, rainforests, vast freshwater swamp forests, and diverse coastal vegetation. The country is also conventionally located concerning solar PV electricity generation potential [56]. With tropical temperatures abound, there is high demand for cooling. Nigeria is faced with environmental problems, such as air pollution, water pollution, lead exposure, and poor waste management. This lowers the potential use of bio-energy [57].

2.2.5. Technical Factors

Technical factors encompass a broad range of issues, such as access to the grid and technology availability. Many Nigerians do not have access to the electricity grid. This could also be associated with the structure of the electricity infrastructure and the existing energy policy [56].

2.2.6. Socio-Cultural Factors

Socio-cultural factors include participation, social cohesion, and identity within a community or an estate. An organization or a housing estate consists of people from different backgrounds, cultures, orientations, and understandings of this concept. CREI formation may encounter local opposition if the community members do not understand what they stand to benefit overall [58]. Furthermore, local support and acceptance within a community towards CREI are related to the degree of ownership and expectancy of receiving a fair and equal share of its benefits. When there is an unequal allocation of profits, hostility might set, resulting in poor local support of the projects [50].

One must also stress that the degree of social cohesion and level of trust within the estates under study in this research can influence the local perception, acceptance, support, and formation of CREIs.

2.2.7. Availability of Expertise and High Costs

Renewable energy technology (RET) suppliers and installers play an essential role for communities or people to adopt RET and CREIs in Nigeria. The main barrier concerns the excessive cost of solar PV due to the lack of local manufacturers and incentives for importing agents [30].

Policymakers can pay attention to this aspect to lower the barriers to small-scale generators entering the supply market. Furthermore, the current effort of the Lagos government and Siemens Nigeria in training interested experts in this field is not enough to provide sufficient skilled local installers with this technology soon. The Lagos State Energy Academy has set a target of training 100,000 energy experts, including installers, power engineers, and energy managers, to be reached in 2030 [59].

In other West-African countries, such as Ghana, local microgrids using solar PV have been installed and tested, showing a relatively high levelized cost of electricity. Compared with a common end-user tariff, end-users of this PV-based mini-grid system end up paying more, even under full-grant finance conditions (e.g., for maintenance of the electricity system). This means that solar PV-based systems are comparatively expensive compared to central-grid-supplied electricity. More in general, cost is considered a significant barrier to the diffusion of renewable energy in African countries, especially with regard to high initial capital costs [33,36].

3. Research Design and Methodology

This research presents a case study of two housing estates in Lagos State, Nigeria, and uses a mixed methods approach based on both qualitative and quantitative data. The two housing estates—Ojokoro and Sunshine—have a combined 400 housing units, with all of the households having fossil-fuel-powered generators in use when electricity supply from the central electricity grid fails. The adoption of RET and willingness to collaborate and operate an integrated community energy system within the estates would potentially lower the use of fossil fuels from the diesel and petrol generators, the cost of running these generators, and would help increase energy security.

To understand the role of the supporting structure needed for the CREI framework, a deep dive into the role of each stakeholder was taken using a stakeholder mapping method. Stakeholders were analyzed by mapping their positions and functions, their interrelation with other stakeholders, their opinion about CREIs, and barriers they could experience in the formative years and future operations, using a similar approach applied by Viétor et al. [60], combining a stakeholder analysis with an empirical reflective-theoretical approach. Table 1 presents an overview of the stakeholders involved and Table 2 presents an overview of stakeholder characteristics and their views toward CREI. The tables clarify the stakeholders positions and functions, their interrelation with other stakeholders, and to what extent they are influential in supporting the establishment of CREIs in the estates.

Stakeholders	Stakeholder Function Method of Data Collection		Potential Impact on the Success of CREI	
Nigeria electricity regulatory Commission (NERC)	Overarching Regulator	Virtual Interview	Very high	
Energy Commission of Nigeria (ECN)	Energy Policy Implementation Agency	Face-to-Face Interview	High	
Lagos State Ministry of Energy and Minerals	Policymaker	Face-to-Face Interview	High	
Lagos State Electricity Board (LSEB)	Implementing an agency for the power sector in Lagos	Face-to-Face Interview	Medium	
Lagos Energy Academy (LEA)	Lagos state funded the renewable energy education academy. They provide technician training and professional certification beyond tertiary education.	Face-to-Face Interview	Low	
National Centre for Energy Efficiency and Conservation	Renewable energy research center set up by ECN for the implementation of policy.	Face-to-Face Interview	High	
Community Research and Development Centre (CREDC)	Environmental and Renewable Energy NGO	Virtual Interview	Medium	
Sunshine Estate Chairman	Coordinates activities in the estate	Face-to-Face Interview	Low	
Ecobank Nigeria Limited	One of the commercial banks operating in Nigeria and other West African states. Currently funding renewable energy projects.	Face-to-Face	Medium	

Table 1. Overview of stakeholders involved.

Stakeholders	Stakeholder Function	Method of Data Collection	Potential Impact on the Success of CREI Medium	
Solar Nigeria	A DFID-sponsored project targeting rural communities that are yet to be connected to the grid.	Virtual Interview		
GIZ	German international organization involved in the Nigeria energy support program on rural electrification. Provide advice on enabling frameworks to encourage energy access and work with the private sector in developing micro- and mini-grids.	Virtual Virtu Face-to-Face Interview	Medium	
Eko Electricity distribution company	privatized. It distributes electricity		High	
Sunshine Estate and Ojokoro Estate Residents	Community renewable energy initiative adopters	Focus group discussion and Survey Questionnaire	Very High	

Table 1. Cont.

 Table 2. Stakeholders, their main characteristics, and views towards CREIs.

Stakeholders	Position and Function	Interrelation with Other Stakeholders	Opinions towards CREIs	Barriers Identified by Stakeholders for CREIs Formation	How to Support CREIs?
Nigeria Electricity Regulatory Commission (NERC)	Regulate the activities of the Gencos, Disco, and Transco. They also ensured compliance with market rules and operating guidelines, issuing licenses and permits to market participants.	All activities of the electricity market are regulated by NERC.	Positive	Cost of investment	N/A
Energy Commission of Nigeria (ECN)	It is a statutory body established for the strategic planning and coordination of national policies in the field of energy. ECN is the apex government parastatal empowered to carry out overall energy planning and policy implementation.	Policy formulation and implementation in the energy sector	Positive	Lack of awareness about CREIs.	Help draft power purchase agreement, human resources, help raise awareness
Lagos State Ministry of Energy and Minerals	Policy formulation in the state electricity sector	They coordinate all parastatal related to electricity in Lagos State, such as LSEB and LEA	Positive	Lack of awareness about CREIs.	Help draft power purchase agreement, human resources, help raise awareness
Lagos State Electricity Board (LSEB)	This is the implementing agency for the power sector in Lagos. They aim to maximize power supply through independent power projects (IPPs) and improve public lighting for the citizens of Lagos.	They have held several meetings with the Disco and industrialists on how to make the power situation better in the state.	Positive	Technical knowledge, infrastructure, enlightenment of the citizen, lack of incentives.	Advice, Technical knowledge, system design

Stakeholders	Position and Function	Interrelation with Other Stakeholders	Opinions towards CREIs	Barriers Identified by Stakeholders for CREIs Formation	How to Support CREIs?
Lagos Energy Academy (LEA)	This is an academy dedicated to the training of skilled manpower for the Lagos electricity sector. They provide technician training and professional certification beyond tertiary education.	They work hand in hand with the supervising ministry and parastatal. They are in partnership with Siemens, CET power, and other private sector actors.	Positive	Initial Capital cost, lack of skilled, trained technicians, education, and awareness.	Security of investment by providing certified engineers to install the system, that is, if the grant is provided
National Center for Energy Efficiency and Conservation	This is one of the three energy research centers set up by ECN for the implementation of policy. They train personnel, disseminate information on energy use, and develop standards, codes, and policy guidelines.	They are set up by ECN, and they work together with other stakeholders on areas related to their mandate.	Positive	Low awareness, high cost of installation, and lack of social cohesion amongst people.	Help educate the community
Community Research and Development Center (CREDC)	This is an NGO that focuses on environmental issues and advocates renewable energy.	They are always in contact with the policy maker and regulator.	Positive	Low awareness, high cost of solar batteries, technical amateurs.	Technical support, Help seek funds.
Sunshine Estate Chairman	Coordinate all activities in the estate.	They only interact with the Disco supplying the estate.	Positive	Financial barrier, Awareness.	Mobilization of the estate to support the project
Ecobank Nigeria Limited	Ecobank is one of the commercial banks operating in Nigeria and other West African states. They are currently funding renewable energy projects.	They have provided counterpart funding for Disco during the privatization process	Positive	Maintenance cost, pricing, lack of social cohesion, and the poverty level in the society.	Financial support (loan)
Solar Nigeria	This is a DFID-sponsored project targeting rural communities that are yet to be connected to the grid.	They are in constant interaction with the Federal ministry of power, ECN, and GIZ.	Neutral	Awareness, expensive to startup, lack of understanding of the technology.	Financial support (loan)
GIZ	This German international organization is involved in the Nigeria energy support program, rural electrification. They advise on enabling the framework for energy access and work with the private sector in developing micro- and mini-grids.	They work with NERC on establishing regulations and support mechanisms. They also work with the federal ministry of power on an enabling framework to increase energy access.	Positive	The initial capital and maintenance cost, technical know-how, and low awareness level.	Lagos is not part of their jurisdiction. GIZ is currently working in five states in Nigeria.
Eko Electricity distribution company	This is one of the two distribution companies in Lagos and is newly privatized. They distribute electricity to the Lagos islands and their environs.	All their activities are subjected to NERC approval.	Neutral	Lack of social cohesion because Lagos is a multicultural society, unavailability of experts, awareness, funding, and no enabling structure.	Technical support, if affordable, helps to structure it.

Table 2. Cont.

3.1. Case Selection

With a population of about 165 million [61] and a GDP of 5.09%, 6.66%, and 7.14% for the years 2011, 2012, and 2013 [62], Nigeria is the most populated and fastest-growing economy in Africa [63], is the largest hydropower producer and has the largest oil reserves in the Economic Community of West African States (ECOWAS). The country also has vast potential for renewable energy, including wind and solar. However, the country's electricity sector is unable to meet the increasing electricity demand. Currently, only half of the installed capacity of 8900 MW is available on the national power grid, which prevents 60% of the population from electricity access [64]. In Nigeria, Lagos is the most populated state, using more energy than any other Nigerian state.

3.2. Data Collection

Research into the formation and implementation of community-led energy initiatives in two housing estates in Lagos was conducted by administering questionnaires to the estates' residents (i.e., households; N = 124). In addition, a series of expert interviews were conducted with stakeholders, including the federal state agencies, international development partners, non-governmental organizations, financial institutes, electricity distribution companies, research institutes, and academic institutes (N = 12).

3.2.1. Stakeholder Interviews

A semi-structured questionnaire was used to allow the interviewees to give openended questions with room for clarification and providing argumentation. This type of interviewing fits well with the explorative nature of the present study. The items on the questionnaire related to how stakeholders perceive CREIs in general and view CREI development in Nigeria, particularly to improving energy security. The questions were drafted by the first author under supervision by the co-authors. The questionnaire used in the interviews can be found in Appendix A. The interviews took between 45 and 60 minutes each. To ensure the quality of the data, the interviews were tape-recorded, and a reporter was hired to take notes during the interview and draft interview transcripts afterward. These stakeholders are relevant to providing the information needed for assessing the support structure, barriers, perception, and willingness to take policy recommendations regarding community renewable energy initiatives and the use of distributed generation renewable energy sources.

3.2.2. Household Survey

For the household survey, the estates targeted were mostly inhabited by educated middle and upper-class residents in Lagos, expecting many of them to be equally financially buoyant, increasing the likelihood that they were willing to adopt the CREI concept locally and start a pilot project at their estate.

Sociodemographic characteristics of the sample pertain to: gender (42% female; 58% male); age (20–30: 16%; 31–40: 38%; 41–50: 32%; >51: 14%); marital status (married: 76%; not married: 24%); religion (religious: 100%; Christian: 74%; Islamic: 26%; Traditional: 0%); education level (postgraduate: 36%; graduate: 48%; secondary school: 16%); occupation (student: 16%; artisan: 28%; trader: 24%; white collar worker: 32%; unemployed: 0%); years of residing in the estate (1–5 y: 26%; 6–10 y: 47%; >10 y: 37%). For the quantitative—statistical—analysis, questionnaire items were based on key concepts from the Boon & Dieperink [50] theoretical framework on the development of local renewable energy organizations. Both open- and closed-ended questions were used in the questionnaire.

3.3. Data Analysis

3.3.1. Stakeholder Analysis

A stakeholder analysis was conducted to generate knowledge about the stakeholders potentially relevant to CREI project development to understand their behavior, intentions,

interrelations, and interests; and to assess the influence and resources they bring to bear on decision-making and implementation of CREIs, in line with [65]. The stakeholder analysis consisted of the identification of stakeholder characteristics, views, and experienced barriers vis-à-vis the uptake and integration of CREIs in Nigeria. For the qualitative analysis, the expert interviews were subjected to transcription and content analysis. Interviews from the various stakeholders were then compared against each other based on selected questionnaire items and by listing the twelve interviewees' answers. Stakeholder mapping and analysis were used to learn about the position, interest, function, power, opinion, interrelation, motivation, information, and resource access of stakeholders, which provides additional meaning to the issues raised and perspectives given by the interviewees. Qualitative data analysis was conducted to assess the position, levels of interest, and influence of CREIs, using the stakeholder analysis approach developed by Schmeer [66]. Stakeholder mapping helped to understand the relationship amongst stakeholders, especially their position, influence, and the resource they bring to CREI. In addition, a stakeholder's analysis worksheet (SAW) was used based on a guideline developed by Eden and Ackerman [67]. The SAW can be found in Appendix B.

This study also included stakeholders in the private sector domain, e.g., electricity companies, DSOs, academic and research institutes such as the Lagos Energy Academy, the National Centre for energy efficiency and conservation, and local NGOs involved in community-based energy and housing estate executives. Based on the interviews with stakeholders, a summary report for each stakeholder was made, after which stakeholder mapping was conducted [60]. To improve understanding of their relationship with other stakeholders, their position, influence, and the resources they bring to the CREIs were surveyed as well. The stakeholders were analyzed per interest, urgency, and attitude [67] about RET uptake and approval of CREI practices.

3.3.2. Analysis of the Survey Data

Data analysis of the residents' survey involved descriptive statistical analysis concerning raw and relative frequencies of selected items to gain understanding of the main features of the respondents' perceptions about CREIs and to indicate the characteristics and trends in the data set.

4. Results

The results show that there is no clear understanding of government efforts about RET, with many Nigerians not aware of existing policies on renewable energy. According to the interview with a CREDC representative, a special workshop, seminar, and publicity of the NREEEP are advised to Nigerians to become aware, educated, and enlightened to have a clear understanding and adoption of RET. Thus far, there are only two bottom-up community energy initiatives in the country; one concerns an NGO funded by a grant, and the other was initiated by a corporate organization for corporate social responsibility.

4.1. Stakeholder Analysis

The results of the stakeholder analysis are summarized in Table 3.

Table 3 shows that all of the government organizations surveyed are considered champions, except the research institute, which has a supportive attitude. NGOs such as Solar Nigeria are neutral to this because they are part of a DFID program (Department for International Development, a UK development organization) that mainly focuses on rural areas not connected to the central electricity grid. At the same time, other stakeholders, such as the electricity distribution company, GIZ, ECO Bank, and the housing estates—are supportive of CREIs formation.

On the **dimension of interest**, all of the government organizations show high interest in CREI formation. In contrast, the private sector stakeholders only show medium interest. However, most NGOs express having a low interest in CREI formation and state that this concept is only feasible in areas that are yet to be connected to the grid.

Name of Stakeholder	Type of Stakeholder	Attitude (Champion, Supporter, Neutral, Critic, and Opponent	Power (1 to −1)	Interest (1 to -1)	Urgency (1 to -1)
ECN	Federal government	Champion	1	1	1
NERC	Federal government	Champion	1	1	1
LSME & M	State government	Champion	1	1	1
LSEB	State government	Champion	1	1	1
LEA	Academic Institute	Supporter	0	1	0
Eko Electricity Distribution Company	Electricity supplier	Supporter	0	0	1
ECOBANK	Private sector	Supporter	$^{-1}$	0	$^{-1}$
CREDC	NGO	Supporter	$^{-1}$	-1	0
GIZ	NGO	Supporter	0	$^{-1}$	$^{-1}$
SOLAR NIGERIA	NGO	Neutral	-1	$^{-1}$	-1
Sunshine Housing Estate	End Users	Supporter	0	1	0
Ojokoro Housing Estate	End Users	Supporter	0	1	0
National Centre for Energy Efficiency and Conservation	Research Institute	Supporter	-1	-1	0

Table 3. Stakeholder overview and results of stakeholder analysis (with -1 meaning 'low'; 0 meaning 'moderate' or 'neutral'; and 1 meaning 'high').

CREIs can access the government funding program to bridge the technical and financial barriers if the private sector can set up solar PV microgrids for the estates and hand them over for maintenance and management. However, this would only be possible if the housing estate allows the private sector to do this in the first place. The estate chairman showed high interest because he views CREI-led solar PV microgrids as a realistic alternative to the current unstable centralized electricity supply system.

On the **dimension of power**, all the government organizations show a strong ability to either encourage or discourage the formation of CREIs. This is important because it is assumed that the formation of CREIs is not likely to become successful in the absence of government incentives and regulatory support. On the other hand, the private sector stakeholders and energy suppliers are considered to have a medium degree of power since they cannot support CREIs without government support because they need regulatory approval by the government. Research institutes and NGOs are also perceived to have the same level of power. Even though they have no direct authority over CREIs, their expertise could significantly influence the government's decision-making process regarding CREIs.

All government stakeholders perceive CREI formation as a step-in-the-right direction with high urgency. Energy suppliers also show high urgency because CREI formation will be beneficial to their business. The private sector stakeholders (i.e., the financial institute) show low urgency as they will only finance CREIs that show business cases with a high internal rate of return and short payback period. Research institutes are considered to have medium urgency since they will be valuable in advising on CREIs formation. The Estate chairman showed medium urgency because CREI-led microgrids would require alternative energy sources. All NGOs except (the local) CREDC show a low degree of urgency; Lagos is not part of its jurisdiction due to its metropolitan nature.

4.2. Estate Household Survey Results

The results show the perception of the inhabitants of the estates on CREI formation and operation; of the 124 respondents, 58% expressed a positive perception, whereas 42% expressed a negative one. With proper sensitization and enlightenment on CREI in the estates, the estates' members could potentially collaborate and adopt this concept. The estates are already practicing a few community-owned projects, though not in the energy domain; for example, a community-funded borehole that ensures constant water supply. For the estate members to cooperate in funding the estate-owned water project, they can similarly fund the estate-owned energy system that would guarantee continuous power supply. However, this would only work when supported by the government and experts.

The examination of the knowledge dynamics of the residents of the estates on CREIs and RET showed that 47% viewed CREIs as making energy services accessible for all; 28%

believed that it would help to bridge the energy poverty gap, while 25% were indifferent. Respondents view CREI as not only helping drive sustainable energy transitions but also improving energy security. Over three-quarters of the respondents (79%) hold that CREI can potentially be seen as a solution to Nigeria's energy security problems, while only 12% felt otherwise, and 9% of the respondents were indifferent. The respondents express concerns about the centralized structure of the electricity grid and the lack of regulatory support to CREIs, citizens, and private investment in the electricity sector. Trust is another concern mentioned. This also relates to potential collaboration with other estate residents. About 80% of the respondents expressed concerns about trust in those that would lead the CREI organization in the estate. They expressed that they would only collaborate when trusted persons would oversee the project's operation.

The success of CREI formation in Nigeria would, to some extent, depend on local and government support. The introduction of a new system or concept in society often faces social resistance and opposition. The level of acceptance of CREIs and RET by the residents of the estate is of fundamental importance. However, it requires a support structure as opposition may pose a serious hindrance to CREI formation. Given this premise, the resident's willingness to collaborate to crowdfund a CREI project in the estate showed that 73% of the respondents were open to having the CREIs in their estate, while 16% of the respondents were not interested, and 11% were neutral.

When asked what could potentially motivate households to potentially participate in a CREI with other residents, 12% stated government support, 17% stated it might guarantee constant electricity supply they are willing to collaborate, while 26% stated that it depends on the level of support or cooperation from neighbors because they expect free riders to unjust benefit from collective action (See also Figure 2). The level of social cohesion in the estate can influence CREI formation or project outcome, but that outcome can influence stakeholders. For example, the support of stakeholders in the estate would positively influence the decision of setting up CREI in the estate, and any pushback from the stakeholders would negatively impact the formation of CREI.



Figure 2. Residents' motivation to cooperate with neighbors in a CREI (in percentages).

However, 10% of the respondents stated that it depends on the sincerity of the purpose among those in charge of the project, citing that, in a project such as this; people responsible are liable to give in to corruption, and 16% of the opinion was transparency, the respondents expressed wanting people in charge of the local energy system that can be trusted and held accountable and who can provide regular feedback. Overall, 7% of the respondents stated that they would rather be late adopters because they will only join when CREIs is considered a proven principle that works correctly, and 8% stated they would only cooperate if the government, a philanthropist organization, or a corporate organization can provide the initial investment cost this can motivate them to cooperate while only 5% stated they are willing to cooperate because of climate change. The type of support needed to empower CREIs is presented in Figure 3. Following a question about an opinion, these stakeholders mentioned various support structures, such as awareness raising (e.g., via campaigns) and education to be the most important (with 24% of the respondents). Not surprisingly, educating people on how the CREI concept works, especially in an overcentralized electricity market. The first step towards CREI formation in a country that is yet to liberalize its electricity market is to educate, sensitize them to the concept, and provide a full understanding of the idea and its benefits will influence their decision on the adoption of CREI.



Figure 3. Perceived type of support to empower CREIs (in percentages).

Figure 4 shows the challenges and obstacles that may hinder CREI formation in Nigeria, as stated by the respondents. These pertain to the initial cost of installation (32% of the respondents), lack of social cohesion (20% of the respondents), poor government policy (11% of the respondents), lack of understanding, trust issues, and skepticism (8% of the respondents each), 4% said resistant to change, 10% stated cost of maintenance, while 5% indicated licenses and permits, and just 2% argued that social opposition might post a challenge.



Figure 4. Perceived barriers to the formation of CREIs (in percentages).

To overcome these barriers, households were asked about the factors or intervention strategies needed for CREI and RET's adoption in the estate (See Figure 5). Of the various actions and policies presented, awareness-raising (37%), having a pilot project (16%), provision of finance 13%, flexible regulation to support its formation 13%, government subsidy 10%, grants 6%, and 5% stated provision of technical know-how.



Figure 5. Perceived factors influencing the formation of CREIs (in percentages).

From these findings, it can be deduced that CREIs are considered conditionally capable of increasing access to energy services in Nigeria by the respondents of the two estates and potentially contribute to a sustainable energy future. In addition, the interviewed stakeholders also expressed a favorable opinion about CREIs as a viable approach for increasing energy access, particularly to the rural communities unconnected to the current electricity grid. They argue that CREIs are potentially feasible but will only be sustainable when government regulation supports their development and includes well-trained community members in the formation and maintenance processes. During an interview, a representative of the "Deutsche Gesellschaft fur Internationale Zusammenarbeit" (GIZ)—a German International Development Organization providing tailor-made, cost-efficient, and effective services for sustainable development in developing countries with projects in Nigeria—stated that CREIs could potentially improve energy access, but a dedicated, specialized organization would be needed to drive the process. A representative of the DSO thinks that CREIs could bridge a specific gap in the amount of power available to distribute.

The survey results also show that 79% of the respondents view the development of CREIs as one of the veritable tools for solving the current energy security problems in Nigeria. In addition, 73% of the respondents are willing to cooperate with their neighbors to form a CREI at the estate level. The people's willingness to work with one another to fund CREI shows that if Nigeria's renewable energy policy supports the development of CREI, it will contribute to both a sustainable energy future and energy security.

Currently, DFID and other international organizations have initiated some programs that focus on electrifying rural communities without grid connection, one of which is the solar Nigeria program. The Nigeria Bank of Industries, on 14 October 2015, inaugurated a 24 kW micro-grid solar solution in two communities in Osun State, in Southwest Nigeria. To address the rural-urban migration due to the inadequate electricity supply and improve affordable power through renewable energy technologies [68]. The Lagos State government is building capacity to bridge the gap in a skilled workforce by training and re-training experts in the power sector through the Lagos energy academy. The availability of international electricity companies in Nigeria, such as Schneider Electric, General Electric, and Siemens, might improve technical know-how and may indirectly promote the adoption of RET and the formation of CREIs in Nigeria.

The Energy Commission of Nigeria has already set up platforms to encourage Nigerian citizens to adopt solar energy, such as Nigeria's solar energy society, which is associated with solar energy promotion. Furthermore, the National Electricity Regulatory Commission stated that CREI would allow quicker access to energy, and the rural electrification fund in view will help develop them. The Lagos State Electricity Board believes estates in Lagos State could cooperate to improve energy access.

5. Discussion

The present study provides information on the perceptions that the estate residents have about CREIs as a new perspective for solving energy security challenges in Nigeria. Within the body of literature on community energy, research into how citizen collectives can contribute to improving energy security is a research avenue that has only recently taken off [69]. The present study contributes to this. Whereas the previous study adopted a modeling approach [69], the present study delivers empirical insights from both an in-depth case study and household survey data.

The results from the present study reveal mixed support for CREI formation from both stakeholders and estate residents. Although the latter expressed some benefits, there are also considerable perceived barriers. For example, the constraints of the domestic electricity market on the centralized system of the political structure do not provide the right incentives to support the formation of CREIs. To do so, there is a dire need to have the right policy support incentives and schemes in place. This is also shown in the survey results, with the residents mentioning that interventions such as low-interest government loans (19%), start-up grants (8%), subsidies (8%), regulatory support (5%), and other measures would probably not work without at least some government support, and that awareness raising and education (24%) would be necessary to empower CREIs. Furthermore, they also claimed that the lack thereof (i.e., in terms of 'poor policy') was a key barrier.

Studies have shown that energy policy support schemes and financial-economic incentives, such as the feed-in tariff and net-metering have driven the popularity of CREIs in Western Europe [70]. This is in line with the results of the present study, as a lack of policy support or incentives for CREI formation in Nigeria is perceived as an impediment to its development (as revealed in the stakeholder analysis and the residents' survey). This result can also be viewed in light of renewable energy source adoption, which is in line with more countries in the region [33]. This also applies to the results of the residents' survey, where the upfront cost is considered the main barrier (32%), and reveals the importance of such problems because they have great implications for (future) CREI's business model. Given that solar PV microgrids can still be considered an expensive alternative to common central-grid-supplied electricity [36], along with the other barriers observed, one cannot expect that viable business models can be generated by CREIs that would work in the absence of at least one supportive financial incentive implemented by the government [43]. In the absence of these incentives, viable business cases are not likely to be generated, and CREIs are even less likely to emerge (with initiators not daring to take the risk), let alone private sector investment.

The present study also shows that stakeholders argue that CREIs would be in need of support and empowerment, not just in terms of favorable policy but also in terms of capacity building, as indicated in the GIZ interview. This indicates the importance of intermediaries supporting CREI formation, organizational operation, brokering contracts, and lowering transaction costs [43], as also observed in countries that witnessed the emergence of community energy in recent years [16,71,72]. Another observation indicated the importance of social cohesion and trust, as witnessed in the residents' survey, where the former (20%) and the latter (8%) came across as key barriers to CREI formation. This underlines the importance of these social characteristics of local communities in which a CREI is formed, and is in line with results from other studies [13,73].

Energy transition geared towards generating energy through a decentralized form of governance that empowers local actors is gaining momentum in various countries across the world [45,74]. At the same time, Nigeria still heavily depends on crude oil for energy. The government has promised to unbundle the power sector to include microgeneration. This is contained in a new legislative bill proposed in June 2021 [75]. The form of local action using a CREI model is one of the strategies that the Government of Nigeria can potentially adopt when it is right to unbundle this critical economic sector. Community renewable energy generation and supply that are either wholly owned or managed by the

members who contribute financially to its development or organized in a partnership with utility companies helping to address the country's energy poverty and security issues.

Increasing the production and consumption of energy from renewable energy sources is fundamental to sustainable development. Being a signatory to the SDGs, Nigeria can domesticate these goals, particularly SDG7, by supporting and promoting community renewable energy development in housing estates in Nigeria. SDG7 clearly outlines the UN's ambitions to ensure access to affordable and clean energy [76].

The continuous dependence on unsustainable energy sources and the adverse effects of greenhouse gas emissions means that switching to renewable energy is essential. The service provided by energy is vital to economic growth, progress, development, and the eradication of poverty in any country aspiring to meet the UN's SDG7 [77].

Reforming energy policy while considering the need to transition to a low-carbon economy, energy security, and citizens' participation in democratizing Nigeria's energy systems could contribute to social and environmental justice. Social and environmental justice is related to the equitable distribution of wealth in society [77]. Energy from community-led projects is one approach that can contribute to environmentally and socially just energy transitions and to SDG7 [42].

Therefore, governments of developing countries are advised to be strategic when addressing energy security issues by using CREIs as one of the possible approaches to green energy systems and improving energy security. For CREIs to be successfully implemented, national governments must ensure that the right people receive the correct information and the proper support at the appropriate time.

6. Conclusions

This paper started with the research question: In which ways and under which conditions can Community Renewable Energy Initiatives (CREI) contribute to the use of renewable energy sources and improve energy security in Nigeria as perceived by selected stakeholders and households? The present study used a mixed method research approach to answer the research question and did so by combining a stakeholder analysis and a household survey.

The former showed that government actors have a fairly positive attitude towards CREI formation. Government actors also acknowledge the high urgency of the situation and are considered to be the most capable of potentially supporting the formation. Most other stakeholders are supportive of CREI formation but see less urgency and were revealed to have less interest. The government is also perceived as having the most powerful role. The private sector depends on it because it requires (legal) approval. Furthermore, CREIs depend on the government because they require a favorable regulatory framework; this is currently not the case in Nigeria.

The household survey revealed how residential sector end-consumers perceived local CREI formation. First, there is a mixed view of CREIs, with 58% of the respondents having a positive view and 42% having a negative view. Nonetheless, 79% expressed that a CREI-led solar PV microgrid could be a possible solution toward achieving increased levels of energy security, and 73% expressed openness to having a CREI present on their estate. The survey also revealed that the estate members are experienced in running community-owned projects, yet not in the energy domain. On those occasions, the housing estate did require technical, financial, as well as government support. Having this perspective, the residents expressed viewing the currently poor regulatory framework as a big challenge to potential CREI formation. An additional barrier would be the lack of trust in other community members, especially concerning staffing, leadership, and running the operations of the CREI organization. Nonetheless, the support of neighbors was seen as an important motivation to participate in a local CREI, as was the wish for a constant electricity supply. When asked about perceived barriers to CREI formation, residents mentioned upfront costs, lack of (social) cohesion in the estate, and poor government policy (unsupportive to RET adoption, microgrid development, and formation of CREIs). Perceived factors to encourage

CREI formation pertain to awareness raising, (financial and technical) project support, and the desire to have a supportive regulatory framework available.

In summary, the estate residents consider CREIs as conditionally capable of increasing access to energy services and improving energy security in Nigeria. However, CREIs are only viewed as a viable option once government regulation is in place to support their development and includes well-trained community members in the formation of its strategic and operational processes. Moreover, the residents also expressed that CREIs will most likely best succeed in remote rural areas.

6.1. Limitations and Suggestions for Future Research

The households and housing estates in the Lagos State of the present study are, to a limited extent, representative of the population of Nigeria. This also holds for a number of the stakeholders consulted. Extrapolating the results of the present study to the general population of Nigeria would be misleading. However, given the explorative nature of the study and the investigation of the openness, perceptions, and potential acceptance of novel energy and social innovation that could potentially alleviate energy security problems, the results are of importance, particularly regarding the social and policy dimensions of energy issues, and about solar PV microgrids, which had previously only been studied from a technical and financial perspective in the West African region (e.g., [36]).

Future research can address the social acceptance of such energy systems and social innovations such as CREIs in more detail or use a citizen sample with greater representativeness of Nigeria. Furthermore, due to logistical constraints, not all of the relevant housing estates and stakeholders in the electricity sector were covered in the present study. This opens a window of opportunity for further research on community energy initiatives in Nigeria that would cover the thirty-six states that were not included in the present study.

An important limitation of the present study concerned the fact that the CREI concept is new to Nigeria, affirming that this is the first study using this concept in Nigeria, according to one of the stakeholders interviewed. This made the process of data collection tedious, as the concept constantly required explanation and illustration to the interviewees and survey respondents. Future researchers are advised to consider replicating the present study a few years from now when CREIs are possibly more present in Nigeria. Another issue pertained to not being able to interview solar PV panel distributors in Nigeria, many of whom believe that the present study was only conducted for market research purposes of their competitors, making them feel uncomfortable divulging information for fear of losing market distribution potential. Future researchers might consider using dedicated tactics to convince solar PV asset suppliers to participate in the study.

Finally, replication of the present study is proposed for future researchers in comparable developing countries where community energy has not yet landed. The present study did not focus on technical and financial-economic details of CREI-managed solar PV microgrids. Future research, taking a more multidisciplinary approach, could complement the research approach central to the present study with technical and economic details, which would also allow for highlighting the importance of business models or presenting and discussing a business case.

6.2. Policy Recommendations

For CREIs to be successful in Nigeria, it is suggested that the government enacts an appropriate renewable energy policy with fiscal incentives, such as feed-in tariffs, to attract investors and communities that are willing to cooperatively fund their independent energy systems, such as the two estates presented in this paper. NGOs and government agencies must also design a robust education and awareness program for communities to adopt this concept and a few pilot projects to demonstrate how CREIs work. Policymakers are also advised to support or set up intermediary support (schemes) to empower CREI initiatives. With the stakeholders performing their functions and all of the necessary factors accounted

for, CREI formation can potentially become successful in Nigeria and help enhance its greening of energy supply and improve the energy security.

Author Contributions: Conceptualization, O.S.O., T.H. and F.C.; methodology, O.S.O., T.H. and F.C.; validation, O.S.O., T.H., and F.C.; formal analysis, O.S.O., T.H.; investigation, O.S.O.; data curation, F.C.; writing—original draft preparation, O.S.O. and T.H.; writing—review and editing, O.S.O. and T.H.; visualization, O.S.O.; supervision, F.C. and T.H. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: The authors confirmed that the data supporting the findings of this study are primarily collected and analyzed by the authors and are available upon request.

Acknowledgments: The authors would like to extend their appreciation to the following organizations: Energy Commission of Nigeria, Lagos Energy Academy, Lagos State Electricity Board, Nigeria Electricity Regulatory Commission, Nigeria Energy Support Program, Deutsche Gesellschaft Fur International Zusammenarbeit, Eko Electricity Distribution Company, and Federal Ministry of power, for providing personnel who participated in interviews. he authors would also like to thank the householders who participated in the survey. In addition, the authors would like to thank the Editor of the Special Issue and two independent reviewers for their comments on a previous version of this paper.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A. Questionnaire Used in Stakeholder Expert Interviews

- 1. What is your role within this organization?
- 2. How did your company get involved in the NESP?
- 3. Have you heard of the Local energy initiative?
- 4. What is your organization's opinion on the concept of local energy initiatives?
- 5. Do you think they are feasible and sustainable in Nigeria?
- 6. Do you know any existing LEIs or community-based energy systems in Nigeria?
- 7. If yes, could you shed more light on it perhaps give us a contact
- 8. Do you think the Local energy initiative if adopted could improve energy access in Nigeria?
- 9. Do you think Nigerians can adopt local energy initiatives? If yes, can you explain further, and if not could you suggest some of the challenges and barriers to the adoption of Local energy initiatives and renewable energy in Nigeria?
- 10. One of the core objectives of NESP is to promote renewable energy investment in Nigeria, what is the current adoption level of renewable energy technology in Nigeria?
- 11. In your opinion what are the barriers to renewable energy technology (RET) in Nigeria?
- 12. In your opinion, what strategy do you think can be put in place as a sort of motivation for Nigerians to adopt the local energy initiative and RET?
- 13. What support can your organization give to potential volunteers or communities who are willing to form an energy cooperative or local energy initiative?
- 14. Support in what form?
- 15. Would you add something that might be considered important for this study?
- 16. Do you know any other organization that performs the same role as yours?

Appendix B. Stakeholder Analysis Worksheet

Table A1. Stakeholder analysis worksheet.

		Attributes Used	in the Research		
		Level of Attitude			
Definition	Champion	Supporter	Neutral	Critic	Opponent
The level of stakeholder support opposes, or neutral about CREIs	Stakeholders that give significant support to CREIs	Stakeholders that give less significant support for CREIs	Stakeholders that neither oppose nor support CREIs	Stakeholders that criticize the projects and operations of CREIs but do not oppose CREIs.	Stakeholders that oppose the projec and operations of CREIs
	Leve	el of power, Level of In	terest and Level of Urg	gency	
Definition	Hig	h (1)	Mediu	ım (0)	Low (-1)
1. Level of Power					
The ability of the stakeholder to affect the CREIs	decisions regardir	Stakeholders can make overreaching decisions regarding the formation of CREI in their jurisdiction.		The stakeholder is one of the several authorities that can make decisions on the formation of CREI. The stakeholders also depend to some extent on other stakeholders.	
2. Level of Interest					
The stakeholder's interest in the formation of CREIs. The disadvantage and/or advantage of CREI to stakeholders	interest or advar	When the stakeholders have a high interest or advantage through the formation of CREIs		When the stakeholders have a medium interest or benefit from CREIs	
3. Level of Urgency					
The urgency of a stakeholder to successfully form CREIs	in ensuring the form any opposition from	upport is very urgent nation of CREIs, and a stakeholders would formation of CREIs	The support of stak urgent to the formati absence of its supp formation	on of CREIs, and the port will affect the	The stakeholder's support is less urgent, and a lack of support will no significantly affec the formation of CREIs

References

- 1. UNEP. Developing Effective Off-Grid Lighting Policy—Guidance Note for Governments in Africa. Available online: https://united4efficiency.org/resources/developing-effective-off-grid-lighting-policy-guidance-note-governments-africa/ (accessed on 4 September 2022).
- 2. IEA. Energy Security; Ensuring the Uninterrupted Availability of Energy Sources at an Affordable Price. Available online: https://www.iea.org/areas-of-work/ensuring-energy-security (accessed on 4 September 2022).
- Nawaz, S.M.N.; Alvi, S. Energy security for socio-economic and environmental sustainability in Pakistan. *Heliyon* 2018, 4, e00854. [CrossRef] [PubMed]
- 4. Osueke, C.; Ezugwu, C. Study of Nigeria energy resources and its consumption. Int. J. Sci. Eng. Res. 2011, 2, 121–130.
- 5. Azeez, W. Nigerians Hit by Power Outage as National Grid Collapses Again. Available online: https://www.thecable.ng/ nigerians-hit-by-power-outage-as-national-grid-collapses-again (accessed on 17 March 2022).
- 6. Olajide, M.; Adeleke, A. Unemployment in Nigeria: Implication for Youths' Advancement and National Development. *Ilorin J. Adm. Dev. IJAD* **2019**, *5*, 71–77.

- 7. Usman, Z.G.; Abbasoglu, S.; Ersoy, N.T.; Fahrioglu, M. Transforming the Nigerian power sector for sustainable development. *Energy Policy* **2015**, *87*, 429–437. [CrossRef]
- 8. Koirala, B.P.; Koliou, E.; Friege, J.; Hakvoort, R.A.; Herder, P.M. Energetic communities for community energy: A review of key issues and trends shaping integrated community energy systems. *Renew. Sustain. Energy Rev.* 2016, *56*, 722–744. [CrossRef]
- Acosta, C.; Ortega, M.; Bunsen, T.; Koirala, B.P.; Ghorbani, A. Facilitating energy transition through energy commons: An application of socio-ecological systems framework for integrated community energy systems. *Sustainability* 2018, 10, 366. [CrossRef]
- Brummer, V. Community energy-benefits and barriers: A comparative literature review of Community Energy in the UK, Germany and the USA, the benefits it provides for society and the barriers it faces. *Renew. Sustain. Energy Rev.* 2018, 94, 187–196. [CrossRef]
- 11. Bomberg, E.; McEwen, N. Mobilizing community energy. Energy Policy 2012, 51, 435–444. [CrossRef]
- 12. Coenen, F.; Hoppe, T. Introduction. In *Renewable Energy Communities and the Low Carbon Energy Transition in Europe*; Palgrave MacMillan: Cham, Switzerland, 2021; pp. 1–22.
- 13. Seyfang, G.; Hielscher, S.; Hargreaves, T.; Martiskainen, M.; Smith, A. A grassroots sustainable energy niche? Reflections on community energy in the UK. *Environ. Innov. Soc. Transit.* **2014**, *13*, 21–44. [CrossRef]
- 14. Hewitt, R.J.; Bradley, N.; Baggio Compagnucci, A.; Barlagne, C.; Ceglarz, A.; Cremades, R.; McKeen, M.; Otto, I.M.; Slee, B. Social innovation in community energy in Europe: A review of the evidence. *Front. Energy Res.* **2019**, *7*, 31. [CrossRef]
- 15. Hoppe, T.; de Vries, G. Social Innovation and the Energy Transition. Sustainability 2019, 11, 141. [CrossRef]
- Seyfang, G.; Park, J.J.; Smith, A. A thousand flowers blooming? An examination of community energy in the UK. *Energy Policy* 2013, *61*, 977–989. [CrossRef]
- 17. Saintier, S. Community Energy Companies in the UK: A Potential Model for Sustainable Development in "Local" Energy? Sustainability 2017, 9, 1325. [CrossRef]
- 18. Lauber, V. Switching to Renewable Power: A Framework for the 21st Century; Routledge: London, UK, 2012.
- 19. Rogers, J.C.; Simmons, E.A.; Convery, I.; Weatherall, A. Public perceptions of opportunities for community-based renewable energy projects. *Energy Policy* **2008**, *36*, 4217–4226. [CrossRef]
- Milchram, C.; Künneke, R.; Doorn, N.; van de Kaa, G.; Hillerbrand, R. Designing for justice in electricity systems: A comparison of smart grid experiments in the Netherlands. *Energy Policy* 2020, 147, 111720. [CrossRef]
- 21. Ruggiero, S.; Martiskainen, M.; Onkila, T. Understanding the scaling-up of community energy niches through strategic niche management theory: Insights from Finland. *J. Clean. Prod.* **2018**, 170, 581–590. [CrossRef]
- 22. Lammers, I.; Heldeweg, M.A. Smart design rules for smart grids: Analysing local smart grid development through an empiricolegal institutional lens. *Energy Sustain. Soc.* 2016, *6*, 36. [CrossRef]
- 23. Lammers, I.; Diestelmeier, L. Experimenting with Law and Governance for Decentralized Electricity Systems: Adjusting Regulation to Reality? *Sustainability* **2017**, *9*, 212. [CrossRef]
- 24. Avelino, F.; Wittmayer, J.M.; Pel, B.; Weaver, P.; Dumitru, A.; Haxeltine, A.; Kemp, R.; Jørgensen, M.S.; Bauler, T.; Ruijsink, S. Transformative social innovation and (dis) empowerment. *Technol. Forecast. Soc. Chang.* **2019**, *145*, 195–206. [CrossRef]
- Omoju, O. The Future of Nigeria Power Sector Post Reform. The Young Scientist of the Lindau Nobel Laureate Meeting. 2014. Available online: http://blog.lindau-nobel.org/the-future-of-nigerias-power-sector-post-reform (accessed on 20 September 2021).
- 26. Adegbite, A.; Fajemirokun, J.; Mustapha, A. Nigeria: The Energy Regulation and Markets Reviews. 2022. Available online: https://thelawreviews.co.uk/title/the-energy-regulation-and-markets-review/nigeria (accessed on 22 May 2022).
- Othuke, O.; Achinike, C. Nigeria: Energy Laws and Regulations 2022. Available online: https://www.globallegalinsights.com/ practice-areas/energy-laws-and-regulations/nigeria (accessed on 22 May 2022).
- Board, P.E. Review of Power Sector Privatisation Overdue. Punch 2021. Available online: https://punchng.com/review-of-power-sector-privatisation-overdue/ (accessed on 22 May 2022).
- 29. Colell, A.D.; Pohlmann, A. Community Energy Projects Redefining Energy Distribution Systems: Examples from Berlin and Hamburg. *Local Energy Auton. Spaces Scales Politics* **2019**, *1*, 213–237.
- Kolawole, S. Playing Politics with Power Sector. Energy Central News. 2020. Available online: https://energycentral.com/news/ playing-politics-power-sector (accessed on 4 September 2022).
- Haller, C. Siemens and Nigerian Government Signed Implementation Agreement for Electrification Roadmap; Siemens Press: Munich, Germany, 2019; Available online: https://assets.new.siemens.com/siemens/assets/api/uuid:451ebc5d-dafb-4248-a4bb-b7cbaf9 dcae1/PR201907235307EN (accessed on 23 October 2021).
- 32. Norouzi, F.; Hoppe, T.; Elizondo, L.R.; Bauer, P. A review of socio-technical barriers to Smart Microgrid development. *Renew. Sustain. Energy Rev.* 2022, 167, 112674. [CrossRef]
- Ouedraogo, N.S. Opportunities, barriers and issues with renewable energy development in Africa: A comprehensible review. *Curr. Sustain. /Renew. Energy Rep.* 2019, 6, 52–60. [CrossRef]
- Madriz-Vargas, R.; Bruce, A.; Watt, M. A Review of factors influencing the success of community renewable energy minigrids in developing countries. In Proceedings of the 2015 Asia-Pacific Solar Research Conference, Brisbane, Australia, 8–9 December 2015; pp. 8–9.

- 35. Coenen, F.; Hoppe, T. *D3.1 Report on Specific Tools of Supplying REScoops in Europe*; University of Twente: Enschede/Delft, The Netherlands, 2016; pp. 1–82.
- Muyiwa, A.; Quansah, D.; Agelin, C.M.; Paul, S.S. Multipurpose renewable energy resources based hybrid energy system for remote community in Northern Ghana. J. Sustain. Energy Technol. Assess 2017, 10, 161–170.
- Walker, G. What are the barriers and incentives for community-owned means of energy production and use? *Energy Policy* 2008, 36, 4401–4405. [CrossRef]
- 38. Sloot, D.; Jans, L.; Steg, L. In it for the money, the environment, or the community? Motives for being involved in community energy initiatives. *Glob. Environ. Chang.* 2019, *57*, 101936. [CrossRef]
- Warbroek, B.; Hoppe, T.; Bressers, H.; Coenen, F. Testing the social, organizational, and governance factors for success in local low carbon energy initiatives. *Energy Res. Soc. Sci.* 2019, 58, 101269. [CrossRef]
- Forrest, N.; Wiek, A. Learning from success—Toward evidence-informed sustainability transitions in communities. *Environ. Innov. Soc. Transit.* 2014, 12, 66–88. [CrossRef]
- Creamer, E.; Eadson, W.; van Veelen, B.; Pinker, A.; Tingey, M.; Braunholtz-Speight, T.; Markantoni, M.; Foden, M.; Lacey-Barnacle, M. Community energy: Entanglements of community, state, and private sector. *Geogr. Compass* 2018, 12, e12378. [CrossRef]
- 42. Jenkins, K.; McCauley, D.; Heffron, R.; Stephan, H.; Rehner, R. Energy justice: A conceptual review. *Energy Res. Soc. Sci.* 2016, 11, 174–182. [CrossRef]
- Nolden, C.; Barnes, J.; Nicholls, J. Community energy business model evolution: A review of solar photovoltaic developments in England. *Renew. Sustain. Energy Rev.* 2020, 122, 109722. [CrossRef]
- 44. Hoppe, T.; Graf, A.; Warbroek, B.; Lammers, I.; Lepping, I. Local governments supporting local energy initiatives; Lessons from the best practices of Saerbeck (Germany) and Lochem (The Netherlands). *Sustainability* **2015**, *7*, 1900–1931. [CrossRef]
- Wierling, A.; Schwanitz, V.; Zeiß, J.; Bout, C.; Candelise, C.; Gilcrease, W.; Gregg, J. Statistical Evidence on the Role of Energy Cooperatives for the Energy Transition in European Countries. *Sustainability* 2018, *10*, 3339. [CrossRef]
- 46. Oteman, M.; Wiering, M.; Helderman, J.-K. The institutional space of community initiatives for renewable energy: A comparative case study of the Netherlands, Germany and Denmark. *Energy Sustain. Soc.* **2014**, *4*, 11. [CrossRef]
- 47. Kooij, H.-J.; Lagendijk, A.; Oteman, M. Who Beats the Dutch Tax Department? Tracing 20 Years of Niche–Regime Interactions on Collective Solar PV Production in The Netherlands. *Sustainability* **2018**, *10*, 2807. [CrossRef]
- Joshi, G.; Yenneti, K. Community solar energy initiatives in India: A pathway for addressing energy poverty and sustainability? Energy Build. 2020, 210, 109736. [CrossRef]
- 49. International Cooperative Alliance. What is a Co-Operative? Available online: http://ica.coop/en/what-co-operative (accessed on 22 May 2022).
- Boon, F.P.; Dieperink, C. Local civil society based renewable energy organisations in The Netherlands: Exploring the factors that stimulate their emergence and development. *Energy Policy* 2014, 69, 297–307. [CrossRef]
- 51. Hicks, J.; Ison, N. An exploration of the boundaries of 'community'in community renewable energy projects: Navigating between motivations and context. *Energy Policy* **2018**, *113*, 523–534. [CrossRef]
- 52. Papa, R.; Gargiulo, C.; Zucaro, F. Urban systems and energy consumptions: A critical approach. In *TeMA: Journal of Land Use, Mobility and Environment*; University of Naples Federico II: Naples, Italy, 2014.
- 53. Brummer, V. Of expertise, social capital, and democracy: Assessing the organizational governance and decision-making in German Renewable Energy Cooperatives. *Energy Res. Soc. Sci.* **2018**, *37*, 111–121. [CrossRef]
- 54. Agbor, D. Corruption in the Power Sector. Available online: https://www.nigeriaelectricityhub.com/2020/08/04/corruption-in-the-power-sector (accessed on 22 May 2022).
- 55. De Almeida, L.; Cappelli, V.; Klausmann, N.; van Soest, H. Peer-to-Peer Trading and Energy Community in the Electricity Market: Analysing the Literature on Law and Regulation and Looking Ahead to Future Challenges; International Energy Agency: Paris, France, 2021.
- Nnaji, C.; Uzoma, C.; Chukwu, J. The role of renewable energy resources in poverty alleviation and sustainable development in Nigeria. Cont. J. Soc. Sci. 2010, 3, 31–37.
- Shaaban, M.; Petinrin, J. Renewable energy potentials in Nigeria: Meeting rural energy needs. *Renew. Sustain. Energy Rev.* 2014, 29, 72–84. [CrossRef]
- Devine-Wright, P. Beyond NIMBYism: Towards an integrated framework for understanding public perceptions of wind energy. Wind Energy 2005, 8, 125–139. [CrossRef]
- Trust, D. Lagos Commissions Energy Academy. 2014. Available online: https://dailytrust.com/lagos-commissions-energyacademy (accessed on 22 May 2022).
- 60. Viétor, B.; Hoppe, T.; Clancy, J. Decentralised combined heat and power in the German Ruhr Valley; assessment of factors blocking uptake and integration. *Energy Sustain. Soc.* 2015, *5*, 5. [CrossRef]
- 61. NBS. National Population Estimates; National Bureau of Statistics: Lagos, Nigeria, 2016.
- 62. NBS. Social Statistics Reports; National Bureau of Statistics: Lagos, Nigeria, 2016.
- 63. Magnowski, D. Nigerian Economy Overtakes South Africa's on Rebased GDP. 2014. Available online: https://www.bloomberg. com/news/articles/2014-04-06/nigerian-economy-overtakes-south-africa-s-on-rebased-gdp (accessed on 22 May 2022).

- 64. GIZ. Nigerian Energy Support Program. 2013. Available online: http://www.giz.de/en/worldwide/26374.html (accessed on 20 November 2021).
- 65. Varvasovszky, Z.; Brugha, R. How to do (or not to do) ... A stakeholder analysis. Health Policy Plan. 2000, 15, 338–345. [CrossRef]
- 66. Schmeer, K. Stakeholder analysis guidelines. *Policy Toolkit Strength. Health Sect. Reform* **1999**, *1*, 1–35.
- 67. Eden, C.; Ackermann, F. Problem structuring: On the nature of, and reaching agreement about, goals. *EURO J. Decis. Process.* **2013**, *1*, 7–28. [CrossRef]
- 68. Obasi, S. Bol Sets up Solar Power Plant for Osun Communities. Vanguard Newspaper 2015. Available online: https://www.vanguardngr.com/2015/10/boi-sets-up-solar-power-plant-for-osun-communities/ (accessed on 4 September 2022).
- 69. Fouladvand, J.; Ghorbani, A.; Sarı, Y.; Hoppe, T.; Kunneke, R.; Herder, P. Energy security in community energy systems: An agent-based modelling approach. *J. Clean. Prod.* **2022**, *366*, 132765. [CrossRef]
- 70. Tingey, M.; Webb, J. Governance institutions and prospects for local energy innovation: Laggards and leaders among UK local authorities. *Energ Policy* **2020**, *138*, 111211. [CrossRef]
- 71. Hargreaves, T.; Hielscher, S.; Seyfang, G.; Smith, A. Grassroots innovations in community energy: The role of intermediaries in niche development. *Glob. Environ. Chang.* 2013, 23, 868–880. [CrossRef]
- 72. Warbroek, B.; Hoppe, T.; Coenen, F.; Bressers, H. The role of intermediaries in supporting local low-carbon energy initiatives. *Sustainability* **2018**, *10*, 2450. [CrossRef]
- 73. Walker, G.; Devine-Wright, P.; Hunter, S.; High, H.; Evans, B. Trust and community: Exploring the meanings, contexts and dynamics of community renewable energy. *Energy Policy* **2010**, *38*, 2655–2663. [CrossRef]
- 74. Coenen, F.H.J.M.; Hoppe, T. *Renewable Energy Communities and the Low Carbon Energy Transition in Europe*; Palgrave MacMillan: Cham, Switzerland, 2021.
- Jeremiah, K. Nigeria Considers New Law to Unbundle Power Sector. 2021. Available online: https://guardian.ng/businessservices/nigeria-considers-new-law-to-unbundle-power-sector/ (accessed on 22 May 2022).
- Munro, P.; Van Der Horst, G.; Healy, S. Energy justice for all? Rethinking sustainable development goal 7 through struggles over traditional energy practices in Sierra Leone. *Energy Policy* 2017, 105, 635–641. [CrossRef]
- Schlör, H.; Fischer, W.; Hake, J.-F. Sustainable development, justice and the Atkinson index: Measuring the distributional effects of the German energy transition. *Appl. Energy* 2013, 112, 1493–1499. [CrossRef]